Author’s response to reviews

Title: Comparison of femtosecond laser-assisted corneal intrastromal xenotransplantation and the allotransplantation in rhesus monkeys

Authors:

He Jin (jh930930@qq.com)
Liangping Liu (80534643@qq.com)
Hui Ding (47642051@qq.com)
Miao He (943602453@qq.com)
Chi Zhang (412539893@qq.com)
Xingwu Zhong (zhongxwu@mail.sysu.edu.cn;jinhe930930@hotmail.com)

Version: 2 Date: 28 Oct 2017

Author’s response to reviews:

Dear Editors and Reviewers:

Thank you for your letter and for the reviewers’ comments concerning our manuscript titled “Comparison of femtosecond laser-assisted corneal intrastromal xenotransplantation and the allotransplantation in rhesus monkeys.” (ID: BOPH-D-17-00337). All of the comments are all valuable and very helpful for revising and improving our paper, as well as for guiding our researches. We have studied the comments carefully and have made corrections that we hope will be met with approval.

Yours sincerely,

Xingwu Zhong, MD&PhD
Professor of Ophthalmology, Zhongshan Ophthalmic Center and Sate Key Laboratory of Ophthalmology, Sun Yat-sen University

54SXianlie Road, Guangzhou, China 510060

E-mail: zhongxwu@mail.sysu.edu.cn

Tel: 86-20-87330381

Fax: 86-20-87333190